Author's response to reviews

Title: The value of diffusion-weighted imaging in assessing the ADC changes of tissues adjacent to breast carcinoma

Authors:

Zhang Yili (yilishengli@hotmail.com)
Huang Xiaoyan (hxyfisher@163.com)
Du Hongwen (dwendy@126.com)
Zhang Yun (imagingzhang@126.com)
Chen Xin (chen_x129@163.com)
Wang Peng (yt.wangpeng813@163.com)
Guo Youmin (cj.r.guoyoumin@vip.163.com)

Version: 7 Date: 26 November 2008

Author's response to reviews: see over
Dear Madam/Sir:

I have checked over my manuscript and I made two Minor changes as following:

First, in the results of abstract and the Figure legends of the figure 5 and figure 6, we changed “$\text{mm}^2$” to “$\text{mm}^2$”, which were highlighted in red color.

Second, we added “A, B, C, D” to the individual pictures of figure 2.

We have resubmitted our manuscript and thank you so much for your revision.

Best regards.

Yili Zhang